Author's response to reviews

Title: Association between novel TARDBP mutations and Chinese patients with amyotrophic lateral sclerosis

Authors:

Hui-Ling Xiong (chocolatexhl@163.com)
Jin-Yang Wang (wangjinyanghs@163.com)
Yi-Min Sun (mogu4922@hotmail.com)
Jian-Jun Wu (jungliw@gmail.com)
Yan Chen (chhyann@gmail.com)
Kai Qiao (kateqiao@hotmail.com)
Qiao-Juan Zheng (yrhzqj@yahoo.com.cn)
Gui-Xian Zhao (zhaogx5639@yahoo.com.cn)
Zhi-Ying Wu (zhiyingwu67@yahoo.com)

Version: 5 Date: 7 January 2010

Author's response to reviews: see over
Cover Letter

To the editor:

We have made all formatting changes to the format of our manuscript requested by editorial production team.

Major Revisions:

1. Author list: Qiao-Juan Zheng has been listed 7th in the manuscript.

2. Tables: The table has been created using a Table tool and divided into rows and columns. There are no vertical lines in the table.

3. Typography: The manuscript has been checked and the final correction has been made.

Sincerely yours,

Zhi-Ying Wu, M.D.&Ph.D.